Author’s response to reviews

Title: Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Authors:
Ming-wei Liu (lmw2004210@163.com)
Rui Wei (758437603@qq.com)
Mei-xian Su (Kmsmx24@126.com)
Hui Li (3420220794@qq.com)
Tian-wen Fang (13890801975@163.com)
Wei Zhang (2436730979@qq.com)

Version: 3 Date: 03 Jan 2018

Author’s response to reviews:

The Biomed Central Editorial Team 06 January 2018

Object: MS: BCAM-D-17-00608R1 - Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Ming-wei Liu; Rui Wei; Mei-xian Su; Wei Zhang; Hui Li; Tian-wen Fang

BMC Complementary and Alternative Medicine

Thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your reviewer’s comments.

Responses to the comments of Reviewer

1. Please detail in your Methods section the euthanization method used in your study.
The rats were anaesthetized by intraperitoneal injection of 1% pentobarbital sodium (35 mg/kg body weight) in the study.

2. We had noticed you had included 9 authors in your Authors' contribution section, but only 6 authors are listed on your manuscript. Please clarify.

We have clarified that I listed only 6 authors on our manuscript, and revised our Authors' contribution section as the reviewer indicates.

3. Please upload all figures as separate figure files and remove them from the main manuscript. Please do not include the figure captions within the figure files, but instead place them at the end of the main manuscript in a section titled “Figure legends”.

We have uploaded all figures as separate figure files and removed them from the main manuscript

4. Please delete your supplementary files. They are no longer needed at this stage of production.

We have deleted your supplementary files as the reviewer indicates.

5. We had noticed there is a large amount of overlap between your manuscript and previously published manuscripts. Please revise these paragraphs (especially in the Methods section) in your own words.

We have revised a large amount of overlap between your manuscript and previously published manuscripts as the reviewer indicates.